Author’s response to reviews

Title: Comparison of diagnostic methods for early interproximal caries detection with nearinfrared light transillumination: An in vivo study

Authors:

ismail baltacioglu (ismailbaltacioglu@yahoo.com)

Mehmet Kolsuz (m_eraykolsuz@yahoo.com)

Kaan Orhan (call52@yahoo.com)

Version: 1 Date: 02 Apr 2017

Author’s response to reviews:

Respond to the editors comments:

1) Figure 1 c and d part deleted

2) Level of the text overlap reduced for the listed articles